Author's response to reviews

**Title:** A pilot cohort study of cerebral autoregulation and 2-year neurodevelopmental outcomes in neonates with hypoxic-ischemic encephalopathy who received therapeutic hypothermia

**Authors:**

V Joanna Burton (BurtonJ@kennedykrieger.org)
Gwyndolyn Gerner (GernerG@kennedykrieger.org)
Elizabeth Cristofalo (ecristofalo@jhmi.edu)
Shang-en Chung (schung2@jhmi.edu)
Jacky M Jennings (jjennin1@jhmi.edu)
Charlamaine Parkinson (chenson1@jhmi.edu)
Raymond C Koehler (rkoehler@jhmi.edu)
Raul Chavez-Valdez (rchavez2@jhmi.edu)
Michael V Johnston (Johnston@kennedykrieger.org)
Frances J Northington (frances@jhmi.edu)
Jennifer K Lee (jkleez@jhmi.edu)

**Version:** 2  
**Date:** 8 April 2015

**Author's response to reviews:**

Revisions were made to upload the correct version of the manuscript and to add page and line numbers.